Author's response to reviews

Title: Early detection of secondary damage in ipsilateral thalamus after acute infarction at unilateral corona radiata by diffusion tensor imaging and magnetic resonance spectroscopy

Authors:

Chuo Li (lichuo0529@126.com)
Xueying Ling (lingxuey@163.com)
Sirun Liu (tlsr@jnu.edu.cn)
Anding Xu (andingxu@gmail.com)
Yusheng Zhang (yszhang100@yahoo.com.cn)
Shihui Xing (xingsh2006@yahoo.com.cn)
Zhong Pei (peizhong@mail.sysu.edu.cn)
Jinsheng Zeng (zengjs@pub.guangzhou.gd.cn)

Version: 3 Date: 22 April 2011

Author's response to reviews: see over
2011-04-22

Dear editor,

Thank you very much for providing us with a chance to revise our manuscript entitled “Early detection of secondary damage in ipsilateral thalamus after acute infarction at unilateral corona radiata by diffusion tensor imaging and magnetic resonance spectroscopy” (1253214881477765). We are grateful to your constructive suggestions. We have considered these suggestions very carefully and have revised the manuscript accordingly. Enclosed please find the revised manuscript and our point-to-point responses to you. We believe that the revision has improved the manuscript, and look forward to your favorable consideration.

With best regards,

Sincerely yours,

Jinsheng Zeng, MD, PhD,

Department of Neurology and Stroke Centre,

The First Affiliated Hospital, Sun Yat-Sen University,

No. 58 Zhongshan Road 2, Guangzhou, 510080, China.
The following are our point-to-point responses to the editor’s comments.

Responses to the editor:

Comment: Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. Also at 100% the text of Figure 2 is not readable. We recommend you upload a higher resolution version of the figure.

Response: We appreciate editor for the suggestions. In the revised manuscript, we have reordered the figure citation and figures by exchanging Figure 1 and Figure 2, so that they are cited in consecutive order. The changed contents were shown in Page 9 line 1, Page 10 line 8, Page 11 line 11, Page 12 line 14 and Page 23 line 2-19. In addition, we have replaced the old version of Figure 2 by a figure of 600dpi in resolution shown in Figure 1.

Comment: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: Thank you for giving us the opportunity. We have checked our manuscript carefully and found no typographical errors.